Author's response to reviews

Title: Rapid and robust phylotyping of spa t003, a dominant MRSA clone in Luxembourg and other European countries.

Authors:

David M Engelthaler (dengelthaler@tgen.org)
Erin Kelley (ekelley@tgen.org)
Elizabeth M Driebe (edriebe@tgen.org)
Jolene Bowers (jbowers@tgen.org)
Carl F Eberhard (carlfeberhard@gmail.com)
Jesse Trujillo (jst46@email.arizona.edu)
Frederic Decruyenaere (frederic.decruyenaere@lns.etat.lu)
James M Schupp (jschupp@tgen.org)
Joel Mossong (jmossong@AfricaCentre.ac.za)
Paul Keim (paul.keim@nau.edu)
Jos Even (joseven@pt.lu)

Version: 4 Date: 20 June 2013

Author's response to reviews: see over
June 15, 2013

BMC Infectious Diseases Editors,

Thank you for the opportunity to submit a final revised version of our manuscript titled: “Rapid and robust phylotyping of spa t003, a dominant MRSA clone in Luxembourg and other European countries” (MS#2139535009892165). The only revision requested is from Reviewer 2 in regards to labeling on Figure 1. The Figure has been slightly altered to improve clarity and the Figure Legend in the manuscript has been updated based on this revision. We have deposited the DNA sequence data and the tree data in public databases (GenBank and Dryad, respectively), as requested. The accession #s are listed in the Results section of the manuscript. The data will become available upon acceptance of manuscript for publication.

Please let me know if I can provide any further information. Thank you again for your consideration.

Sincerely,

David M. Engelthaler
Director, Programs and Operations